Author's response to reviews

Title: DNA microarray data integration by ortholog gene analysis reveals potential molecular mechanisms of estrogen-dependent growth of human uterine fibroids

Authors:

Tao Wei (weita@lilly.com)
Andrew G Geiser (ageiser@lilly.com)
Hui-Rong Qian (qianhu@lilly.com)
Chen Su (chen_su@lilly.com)
Leah M Helvering (lhelvering@lilly.com)
Nalini H Kulkarni (kulkarninh@lilly.com)
Jianyong Shou (shou@lilly.com)
Mathias N'Cho (n_mathias@lilly.com)
Henry U Bryant (hub30@lilly.com)
Jude E Onyia (joe@lilly.com)

Version: 3 Date: 30 March 2007

Author's response to reviews:

Dear Madam/Sir,

The authors thank you and Dr Segars for taking time reviewing our manuscript again and accept our manuscript for publication. We did not make any major changes except for minor edits to comply with the format guideline. For example,

1. used sentence case in the title, and all headings
2. included the country in the affiliation details.
3. cited websites in the required format.
4. removed any highlighting and tracking in the text
5. added acknowledgement of funding for all authors

Sincerely,

Tao Wei Ph.D
Sr. Research Scientist
Lilly Research Laboratories
Eli Lilly and Co
317-651-9917
weita@lilly.com